Author's response to reviews

Title: Prognostic value of preoperative neutrophil-to-lymphocyte and platelet-to-lymphocyte ratios, and multiphasic renal tomography findings in histological subtypes of renal cell carcinoma

Authors:

Suat Keskin (drsuatkeskin@yahoo.com)
Zeynep Keskin (zkeskin@ymail.com)
Hakan Hakki Taskapu (drtaskapu@hotmail.com)
Havva Kalkan (drhavyaradyoloji2011@gmail.com)
Mehmet Kaynar (mekaynar@gmail.com)
Necdet Poyraz (necdetpoyraz@gmail.com)
Hatice Toy (haticetoy@hotmail.com)

Version: 2
Date: 8 August 2014

Author's response to reviews:

1) Ethics statement was added in line 78,79.
2) Conclusions section was added after discussion.
3) I included line numbers in revised manuscript.
4) Abbreviations, Competing interests, Authors’ contributions and Acknowledgements were added before references.
5) I placed my tables after figure legends